Author's response to reviews

Title: Teriparatide use during an economic crisis: baseline data from the Greek cohort of the Extended Forsteo Observational Study (ExFOS)

Authors:

Kyriakos Aloumanis (Aloumanis_kyriakos@lilly.com)
George Kapetanos (geo.kapetanos@gmail.com)
Nikolaos Bartzis (bartzis_nikolaos@lilly.com)
Vangelis Drossinos (drossinos_vangelis@lilly.com)

Version: 6 Date: 20 May 2015

Author's response to reviews:

As requested, we have updated the Authors’ contributions of the paper to include the specific contributions of each author, and included the following text:

‘KA, NB and VD contributed to data management and interpretation. GK, as National Advisor, also contributed to data acquisition. All authors were involved in drafting the manuscript and replying to reviewers’ comments and have read and approved the final version.’